Author's response to reviews

Title: Seroprevalence of HIV, HBV, HCV and syphilis infections among blood donors at Gondar University Teaching Hospital, Northwest Ethiopia: declining trends over a period of five years

Authors:

Belay Tessema (bt1488@yahoo.com)
Gizachew Yismaw (gizachewyismaw@yahoo.com)
Afework Kassu (afework.gizaw@ucdenver.edu)
Anteneh Amsalu (ant.amsalu@gmail.com)
Andargachew Mulu (andargachewmu@yahoo.com)
Frank Emmrich (Frank.Emmrich@medizin.uni-leipzig.de)
Ulrich Sack (Ulrich.Sack@medizin.uni-leipzig.de)

Version: 6 Date: 27 April 2010

Author's response to reviews: see over
Author's response to reviews

27th April 2010

The BMC Series Editorial Production Team

BMC Infectious Diseases

MS: 5496971193277156 Seroprevalence of HIV, HBV, HCV and syphilis infections among blood donors at Gondar University Teaching Hospital, Northwest Ethiopia: declining trends over a period of five years
Dear Editorial Production Team,

The following formatting changes were made to this revised manuscript.

Major revisions (we require the author to make these changes)

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1]. ' Currently we are unable to find reference 6 cited in the manuscript.

Corrcted.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

1. Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, 'The Title of My Prestigious Publication: A Scientific Study.' should be changed to 'The title of my prestigious publication: a scientific study'.

Corrcted.

2. Abstract: Please remove the keywords from the abstract.

Keywords are now removed.

3. References: please provide the full citation details for reference 15. You can view the BMC reference guide at the following link

Corrcted.

4. References: the reference list should contain up to 30 authors' names for each citation. The term 'et al' should not be used if there are fewer than 30 authors.

Corrcted.

5. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have checked the manuscript for topographical errors and made revisions.
Sincerely,

Belay Tessema